Author's response to reviews

Title: Chronic respiratory disease amongst children presenting to a tertiary paediatric emergency department with acute respiratory illness: study protocol

Authors:

Ben J Drescher (b.drescher@uq.edu.au)
Anne B Chang (anne.chang@menzies.edu.au)
Natalie Phillips (Natalie_Phillips@health.qld.gov.au)
Jason Acworth (Jason_Acworth@health.qld.gov.au)
Julie M Marchant (j.marchant@uq.edu.au)
Theo P Sloots (t.sloots@uq.edu.au)
Kerry-Ann F O'Grady (k.ogrady@uq.edu.au)

Version: 2 Date: 11 May 2013

Author's response to reviews: see over